Author’s response to reviews

Title: Amiodarone-induced reversible and irreversible hepatotoxicity: a case report

Authors:
Toyonobu Tsuda (ttsuda0329@yahoo.co.jp)
Hayato Tada (ht240z@sa3.so-net.ne.jp)
Yoshihiro Tanaka (littlechild0626@gmail.com)
Naoto Nishida (abcing4410@gmail.com)
Taiji Yoshida (tjyd27@gmail.com)
Takeshi Sawada (sawada1046@gmail.com)
Kenji Sakata (kenjis@yu.incl.ne.jp)
Kenshi Hayashi (kenshi@med.kanazawa-u.ac.jp)
Masa-aki Kawashiri (mk@med.kanazawa-u.ac.jp)
Takeru Oyama (takeruoyama@staff.kanazawa-u.ac.jp)
Motoko Sasaki (m8sasaki@med.kanazawa-u.ac.jp)
Nozomu Kurose (k-nozomu@kanazawa-med.ac.jp)
Masakazu Yamagishi (myamagi@med.kanazawa-u.ac.jp)

Version: 2 Date: 19 Feb 2018

Author’s response to reviews:

Answers to Reviewer

Thank you very much for your favorable and constructive comments. We have revised our manuscript according to your suggestions point by point.
Additional revisions are needed.

Add a paragraph at the end of the Introduction that explains why this case report is presented (what is unique and adds to the medical knowledge). Currently there is only a single sentence to introduce the cases.

Our response: We added a sentence at the end of the Introduction section that summarizes the cases highlighting the important messages.

More information is needed:

Each patient should be separately presented. That is - do it under a title of CASE 1, and CASE TWO.

Our response: According to this suggestion, we added headers in our manuscript.

Each case report should include past medical, social, environmental, family and employment history.

What medications was the patient on prior to diagnosis? Did the patient smoke?

Give detailed physical and neurological examination on admission. What was the temperature, pulse, blood pressure and temperature, on admission?

Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc) What were the cardiac examination and cardiac status on admissions and hospital stays?
Our response: We added sentences describing those clinical information in our manuscript as suggested.

Give information about follow-up for at least 6 months for the second case.

Our response: We added a sentence describing follow-up of the second case.

Discussion – add a paragraph at the beginning of the Discussion that summarizes the case and describes what is unique in this case compared to what is available in the literature.

Our response: We added a sentence that summarizes the cases highlighting the important messages.